Author’s response to reviews

Title: Behaviour change strategies for reducing blood pressure-related disease burden: findings from a global implementation research programme

Authors:

David Peiris (dpeiris@georgeinstitute.org)
Simon R Thompson (simon.thompson@ucl.ac.uk)
Andrea Beratarrechea (aberatarrechea@iecs.org.ar)
Maria K Cardenas (maria.cardenas.q@upch.pe)
Francisco Diez-Canseco (fdiezcanseco@upch.pe)
Jane Goudge (jane.goudge@gmail.com)
Joyce Gyrnfi (joyce.gyrnfi@nyumc.org)
Jemima Hoine Kamano (jemhoine@yahoo.com)
Vilma Irazola (virazola@iecs.org.ar)
Claire Johnson (cjohnson@georgeinstitute.org.in)
Andre Kengne (andre.kengne@mrc.ac.za)
Ng Kien Keat (drkkng@hotmail.com)
J. Jaime Miranda (jaime.miranda@upch.pe)
Sailesh Mohan (smohan@phfi.org)
Barbara Mukasa (barbara.mukasa@mildmay.or.ug)
Eleanor Ng (eleanor.ng@phri.ca)
Robby Nieuwlaat (robbynieuwlaat@phri.ca)
Olugbenga Ogedegbe (olugbenga.ogedegbe@nyumc.org)
Bruce Ovbiagele (ovibes@musc.edu)
Jacob Plange-Rhule (jprhule@africaonline.com.gh)
Devarsetty Praveen (dpraveen@georgeinstitute.org.in)
Abdul Salam (asalam@georgeinstitute.org.in)
Margaret Thorogood (m.thorogood@warwick.ac.uk)
Amanda G Thrift (amanda.thrift@monash.edu)
Rajesh Vedanthan (rajesh.vedanthan@mssm.edu)
Salina Waddy (waddysp@ninds.nih.gov)
Jacqui Webster (jwebster@georgeinstitute.org.au)
Ruth Webster (rwebster@georgeinstitute.org.au)
Karen Yeates (yeatesk@queensu.ca)
Khalid Yusoff (yusoff_khalid@yahoo.com)

Version: 2
Date: 22 September 2015

Author’s response to reviews:

Dear Professor Michie,

Thank you for the opportunity to submit a revised version of the manuscript and to provide a response to the reviewers’ comments below. We are fortunate to have reviewers with considerable experience in this area and we appreciate the detailed and helpful feedback they have provided.
The writing group had considerable discussion to facilitate re-working the paper and we have provided a point-by-point response to the comments on the attached document. We look forward to your feedback.

Regards

David Peiris

on behalf of the co-authors.